ŗ



Geophysical Research Letters^{*}

RESEARCH LETTER

10.1029/2022GL100443

Key Points:

- We develop a numerical transport model to understand the long-term fate of CO₂ in gigatonne-scale geologic carbon storage
- CO₂ leakage is dominated by molecular diffusion at inherently slow rates, hardly approaching a meter per several thousands of years
- The long-term potential for CO₂ leakage from a multi-layered system is low, making geologic storage a secure decarbonization technology

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

I. R. Kivi, iman.rahimzadeh@idaea.csic.es

Citation:

Kivi, I. R., Makhnenko, R. Y., Oldenburg, C. M., Rutqvist, J., & Vilarrasa, V. (2022). Multi-layered systems for permanent geologic storage of CO₂ at the gigatonne scale. *Geophysical Research Letters*, 49, e2022GL100443. https://doi. org/10.1029/2022GL100443

Received 26 JUL 2022 Accepted 5 DEC 2022

Author Contributions:

Conceptualization: R. Y. Makhnenko, V. Vilarrasa Data curation: I. R. Kivi, R. Y. Makhnenko, V. Vilarrasa Formal analysis: I. R. Kivi, R. Y. Makhnenko, V. Vilarrasa Funding acquisition: I. R. Kivi, R. Y. Makhnenko, C. M. Oldenburg, J. Rutqvist, V. Vilarrasa Investigation: I. R. Kivi, R. Y. Makhnenko, C. M. Oldenburg, J. Rutqvist, V. Vilarrasa Methodology: I. R. Kivi Resources: V. Vilarrasa Software: I. R. Kivi

© 2022. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Multi-Layered Systems for Permanent Geologic Storage of CO₂ at the Gigatonne Scale

I. R. Kivi^{1,2,3} , R. Y. Makhnenko⁴ , C. M. Oldenburg⁵ , J. Rutqvist⁵ , and V. Vilarrasa³

¹Institute of Environmental Assessment and Water Research, Spanish National Research Council (IDAEA-CSIC), Barcelona, Spain, ²Associated Unit: Hydrogeology Group (UPC-CSIC), Barcelona, Spain, ³Global Change Research Group (GCRG), IMEDEA, CSIC-UIB, Esporles, Spain, ⁴Department of Civil & Environmental Engineering, University of Illinois at Urbana-Champaign, Urbana, IL, USA, ⁵Energy Geosciences Division, Lawrence Berkeley National Laboratory, Berkeley, CA, USA

Abstract The effectiveness of Carbon Capture and Storage (CCS) as an imperative decarbonization technology relies on the sealing capacity of a fine-grained caprock to permanently store CO_2 deep underground. Uncertainties in assessing the caprock sealing capacity increase with the spatial and temporal scales and may delay CCS deployment at the gigatonne scale. We have developed a computationally efficient transport model to capture the dynamics of basin-wide upward CO_2 migration in a multi-layered setting over geological time scales. We find that massive capillary breakthrough and viscous flow of CO_2 , even through pervasively fractured caprocks, are unlikely to occur and compromise the storage security. Potential leakage from the injection reservoir is hampered by repetitive layering of overlying caprocks. This finding agrees with geologic intuition and should be understandable by the public, contributing to the development of climate policies around this technology with increased confidence that CO_2 will be indefinitely contained in the subsurface.

Plain Language Summary Massive and timely deployment of Carbon Capture and Storage (CCS) at the gigatonne scale is a critical component of the majority of pathways toward reaching the Paris Agreement emissions abatement targets to mitigate climate change. Although CCS has been successfully deployed at multiple sites around the world, concerns about long-term containment, especially for gigatonne-scale storage, are causing uncertainty about the ability of geological layers to permanently store CO₂ underground. This uncertainty is delaying the widespread deployment of CCS. This paper focuses on assessing the possibility of basin-wide CO₂ leakage through a typical multi-layered geological setting with a sequence of aquifers and fine-grained caprocks (e.g., shales). Numerical transport models, constrained by hydraulic properties of intact and pervasively fractured caprocks (as the best- and worst-case scenarios, respectively), enable us to draw unambiguous limits on the CO₂ leakage rates over previously unexplored temporal and spatial scales. We show that CO₂ leakage to shallow sediments in both extreme scenarios is quite unlikely and the injected CO₂ will be contained in the subsurface over millions of years. These findings should offer confidence to industrial developers, policy-makers, investors, and most importantly, the public that CCS provides a secure and environmentally sound carbon removal option.

1. Introduction

Rapid worldwide decarbonization is necessary to achieve the Paris Agreement goal of limiting global warming to 1.5° C. A key component of any projected pathway to net-zero CO₂ emissions is carbon capture and permanent storage (Carbon Capture and Storage (CCS)) in deep geological formations (IPCC, 2018; Scott et al., 2012). To become an effective climate-change mitigation strategy, CCS must be deployed basin-wide, which entails a vast scale-up of two orders of magnitude with respect to the currently installed storage capacity (Global CCS Institute, 2021). However, the widespread adoption of the CCS technology has been delayed due, in part, to concerns around the long-term retention of CO₂ underground (Blackford et al., 2014; Monastersky, 2013). CO₂ leakage to shallow sediments would likely degrade potable water resources by releasing heavy metals (Delkhahi et al., 2020; Oldenburg, 2008), and upon reaching the surface, it may kill vegetation (Zhao et al., 2017), animals (Farrar et al., 1995) and people (Hill, 2000) by asphyxia if trapped in topographic lows, as well as challenge the CCS effectiveness in the case of seeping back to the atmosphere at rates exceeding the allowable 0.01% of the stored CO₂ per year (Hepple & Benson, 2005).



Validation: I. R. Kivi

Rutqvist, V. Vilarrasa

Supervision: R. Y. Makhnenko, C. M.

Oldenburg, J. Rutqvist, V. Vilarrasa

Writing – original draft: I. R. Kivi Writing – review & editing: R. Y.

Makhnenko, C. M. Oldenburg, J.

The density contrast of the injected CO_2 relative to resident brine promotes buoyant upward flow of CO_2 (Bachu, 2003). The CO_2 plume can rise several hundreds of meters through permeable sediments in the reservoir during and shortly after the injection, leaving CO_2 bubbles trapped in rock pores by capillary forces (capillary or residual trapping) (Juanes et al., 2010; Krevor et al., 2011). The remaining free-phase CO_2 eventually becomes permanently trapped by dissolution into the resident brine (solubility trapping) and mineral precipitation (mineral trapping) at relatively slow rates (Benson & Cook, 2005; Gunter et al., 2004). The buoyant free-phase CO_2 and the associated leakage hazard may exist for hundreds of thousands of years, necessitating the presence of tight continuous caprock(s) to ensure storage integrity. Competent sealing caprocks, commonly comprising clay-rich formations (referred to here as shales), are characterized in the laboratory by ultra-low intrinsic permeability in the nanodarcy range, small porosity, and high capillary entry pressure of several MPa (Boulin et al., 2017; Heath et al., 2012; Hildenbrand et al., 2002; Makhnenko et al., 2017). These factors narrow down the sources of leakage to improperly sealed wellbores and conductive faults (Alcalde et al., 2018; Song & Zhang, 2013).

Nevertheless, the transport properties of shales are reported to be strongly scale-dependent: the permeability may increase by several orders of magnitude from the lab to the field and regional scales (Hart et al., 2006; Neuzil, 2019). The caprock sealing capacity may also gradually degrade by a number of factors: (a) overpressure creating (micro)fractures or reactivating pre-existing faults (Rutqvist, 2012; Vilarrasa et al., 2014), (b) CO_2 dissolution into water that acidifies the pore fluid and likely promotes mineral dissolution and/or transformation (Gaus et al., 2005; Rezaee et al., 2017), (c) CO_2 adsorption on clay platelets and generation of swelling pressure (Busch et al., 2016), and (d) CO_2 entry in free phase and resulting dehydration of the pore or clay interlayer giving rise to desiccation cracking (Bhuiyan et al., 2020; Espinoza & Santamarina, 2012). These mechanisms have the potential to create pervasive pore-scale heterogeneities that may act as paths of least resistance for CO_2 leakage. An example is the Sleipner storage project, where the CO_2 plume has ascended through eight thin fractured intra-formational claystone barriers (total thickness of 20 m) over the period of 3 years (Cavanagh & Haszeldine, 2014). Furthermore, even an effective capillary seal remains amenable to transport by aqueous CO_2 molecular diffusion. Indeed, diffusive transport, although inherently slow, is ubiquitous and contributes to growing uncertainties in long-term leakage evaluations (Busch et al., 2008; Kivi et al., 2022).

In this study, we address two fundamental questions: how far will the leaking CO_2 front migrate toward the surface over geological time scales, and how is the subsurface CO_2 containment affected by the caprock sealing properties? To address these questions, we substantially advance existing numerical modeling efforts that capture the dynamics of CO_2 leakage but are restricted to decadal to multi-century time scales due to high computational costs (Cavanagh & Haszeldine, 2014; Hou et al., 2012; Zahasky & Benson, 2016). To extend the time and length scales of analysis, we propose here a one-dimensional upscaling model that is arguably capable of dealing with the underpinning physics of leakage at the basin scale and over long-time periods relevant to gigatonne-scale CCS (thousands of years). We estimate the temporal evolution of CO_2 mass distribution across geological strata, finding that the possibility of CO_2 leakage through a multi-layered system of aquifers and caprocks is significantly lower than when a single, even much thicker caprock exists. Such a multi-layered geological setting builds confidence in permanent geologic storage of CO_2 at the gigatonne scale.

2. Materials and Methods

2.1. Description of the CO₂ Leakage Model

To simulate long-term CO_2 storage, we consider continuous CO_2 injection for 30 years at the regional scale through a dense grid of 10-km-spaced wellbores into a deep homogeneous saline aquifer (Figure 1). The planar symmetry of the problem in the *x*- and *y*-directions imposes quasi no-flow boundaries across injection sites, controlling a laterally extensive gravity override CO_2 plume and brine pressure evolutions (Bandilla & Celia, 2017; De Simone et al., 2019; Juanes et al., 2010; Nordbotten et al., 2005; Vilarrasa et al., 2010). Meanwhile, the interfering pressure perturbations of neighboring sites build up a large pressurized region touching the basin margins (Zhou et al., 2010). After the injection ceases, the plume gradually redistributes mainly due to buoyancy and the lateral expansion tapers off, consistent with short-term post-injection observations at the resolution of seismic data at the demonstration projects in Ketzin, Germany (Lüth et al., 2017) and at the CO2CRC Otway site in Australia (Dance et al., 2019). The CO_2 plume shape and pressure distributions in the storage aquifer are expected to approach quasi-equilibrium states a few tens up to hundreds of years after the injection stops (Pawar et al., 2020;



Geophysical Research Letters



Figure 1. Conceptual representation of basin-wide CO_2 injection (not to scale). (a) A bird's eye perspective of CO_2 injection into the reservoir through a dense pattern of wellbores (filled black circles) with a spacing of 10 km. Under buoyancy, the CO_2 plume migrates along the top of the aquifer (blue circular regions; the same color is used for illustration on the *x*-*z* plane). Assumed symmetries in *x*- and *y*-directions, regardless of the interactions between individual CO_2 plumes, simplify the post-injection leakage phenomenon to a 1D (*z*-direction) transport problem. (b) The model comprises a sequence of aquifers and caprocks with representative thicknesses and depths. We extend the succession to the surface, allowing us to impose atmospheric boundary conditions at the top, while the lower boundary immediately above the non-porous crystalline basement is subjected to a no-flow boundary condition.

Zapata et al., 2020). Injection-induced and post-injection hydraulic perturbations can be considered as instantaneous in geologic time, and one may assume that the long-term CO_2 migration will occur almost uniformly in the z-direction. This physically justifiable simplification allows use of a one-dimensional transport model to estimate an upper limit on the CO_2 leakage possibility as the plume can migrate in the post-injection period under natural groundwater flow and the reservoir slope effects (MacMinn et al., 2011). The model geometry comprises an idealized multi-layered system of aquifers and caprocks overlying the low-permeability crystalline basement (Figure 1b).

2.2. Model Parameters

The transport properties of aquifers and caprocks are given in Table S1 in Supporting Information S1. For simplicity, we assign to all aquifers the same set of properties equivalent to those of Berea sandstone as a typical reservoir formation (Vilarrasa et al., 2016). We stipulate two simulation scenarios where all caprocks share either a high or low sealing capacity representing the best-case and worst-case leakage scenarios, respectively. The former scenario adopts the sealing properties of an intact clay-rich shale caprock evaluated in the lab with a typically high-capillary entry pressure p_0 of 2.5 MPa and low intrinsic permeability k of 10^{-20} m² (corresponding to 10 nanodarcy) (Hildenbrand et al., 2002; Makhnenko et al., 2017). The latter case considers $k = 10^{-16}$ m² (corresponding to 0.1 millidarcy), representative of a degraded regional-scale shale permeability (Hart et al., 2006; Neuzil, 2019), and $p_0 = 0.1$ MPa, comparable to that inferred from the Sleipner CO₂ plume migration (Cavanagh & Haszeldine, 2014). Another uncertain parameter is the relative permeability to CO₂ (Kivi et al., 2021), which is hypothesized here to be a power-law function of saturation with an exponent n = 6 for the high-quality seal and n = 3 for the degraded one. We assume that the diffusive transport of aqueous CO₂ through aquifers is established

at the same rate as in bulk water with a coefficient D of 2×10^{-9} m²/s (Tewes & Boury, 2005), whereas it slows down through the tortuous pore network of caprocks with $D = 2 \times 10^{-10}$ m²/s, reaching the upper limit of experimentally determined values (Busch et al., 2008; Rezaeyan et al., 2022).

2.3. Numerical Simulations

We aim at numerically simulating the fate of injected CO₂ in the long term. Therefore, we implicitly account for transient injection-induced effects during the initialization of the post-injection period. Pressure distribution is initially hydrostatic with a uniform dissolved CO₂ concentration of 2.54×10^{-4} kg/kg of low-salinity water. The model is isothermal with a temperature of 40°C for all layers. We inject CO₂ for 30 years at a constant overpressure of 6 MPa to the upper lateral segments of the reservoir while the bottom side permeates water, resembling its displacement by the spreading CO₂ plume or potential leakage to the crystalline basement or at basin margins. We make several trials by varying the water leakage parameter to achieve a CO₂ saturation column and pressure distributions representative of real practices after attaining the aforementioned CO₂ plume stabilization. The long-term CO₂ migration analysis is subjected to constant atmospheric pressure and no-flow conditions at the top and bottom boundaries, respectively. We solve the one-dimensional transport problem for 1 million years (Vilarrasa, 2022) using the finite element code CODE_BRIGHT (Olivella et al., 1996), which was extended to model the CO₂ injection (Vilarrasa et al., 2010, see Section S1 in Supporting Information S1 for the model formulation). We assume that the long-term CO_2 migration is primarily controlled by buoyancy and dissolution into the brine, neglecting potential residual and mineral trapping contributions. The mesh consists of 7,100 quadrilateral elements with a constant thickness of 5 cm for the lowermost 200-m-thick sediments while gradually enlarging toward the surface. The solver discretizes the 10⁶-year simulation period to near 14 million unequal time steps. Sensitivity analyses show that the solution in terms of the spatial and temporal evolutions of the pore pressure and CO_2 saturation and concentration does not change with further refinement of the mesh and time steps. The total CPU time is around 140 hr on a Xeon CPU of speed 2.5 GHz with 32 GB memory.

3. Results

3.1. Best-Case Assessment of the Caprock Sealing Capacity

The combined effects of injection overpressure and buoyancy arising from the CO_2 column height increase the capillary pressure at the base of the caprock to a maximum value of near 2 MPa (Figure 2a). Thus, CO_2 in free phase enters the caprock, but cannot percolate through it. CO_2 remains immobile due to negligible changes in its already very small relative permeability (Figure 2b) and becomes confined to the lowermost portion of the caprock (10 cm). On the contrary, the high relative permeability of the caprock to brine facilitates its flow ahead of CO_2 . CO_2 dissolution into brine leads to a gradual decrease in the excess pressure and, thus, the capillary pressure, initiating an imbibition path after the injection ceases. Brine is imbibed into the pores from the reservoir below until recovering the initial state of saturation. Due to these phenomena, the possibility of a rapid advective leakage of free-phase CO_2 is thoroughly ruled out, and CO_2 transport is exclusively dominated by molecular diffusion. Gravity establishes a hydrostatic brine pressure distribution after 10^5 years that significantly prolongs further CO_2 pressure drop in the reservoir and boosts the diffusive leakage. The dissolved CO_2 is able to propagate through 200 m of overlying formations after 1 million years (Figure 3a). Accordingly, the amount of free-phase CO_2 within the reservoir progressively declines, whereas the storage witnesses a growing contribution of dissolved CO_2 in overlying and underlying formations (Figure 4a).

3.2. Worst-Case Assessment of the Caprock Sealing Capacity

The low-capillary entry pressure and relatively high permeability of the caprocks permit CO_2 to pass through the whole primary caprock and penetrate the secondary one in the course of the injection. CO_2 displaces a considerable portion of the mobile wetting phase out of the pore space and makes a transition to a two-phase flow (Figures 2c and 2e). The relative permeability to CO_2 increases to values as high as 0.6, significantly promoting its advective flux (Figures 2d and 2f). In the early post-injection period, CO_2 redistributes on the imbibition path across the already invaded pore network, including the whole primary caprock and the base of the secondary one, while it continues penetrating the overlying rock layers. Interestingly, the pore fluid of the overlying aquifers and caprocks becomes saturated with dissolved CO_2 (the region with a nearly constant concentration gradient



Geophysical Research Letters



Figure 2. Two-phase flow dynamics of CO₂ leakage. (a) Capillary pressure and (b) CO₂ relative permeability curves with four scanning times $t_1 = 30$ years, $t_2 = 100$ years, $t_3 = 1,000$ years, and $t_4 = 10,000$ years (corresponding to colors gradually shifting from dark to light blue) for a point at the base of the primary caprock in the best-case leakage assessment scenario. Similar curves are plotted for the worst-case leakage assessment scenario at the base of the primary caprock (c and d, respectively) and the base of the secondary caprock (e and f, respectively). The observation points at the basal interfaces of the primary and secondary caprocks are shown by circles and squares, respectively, in the right-hand side of the figure.

proportional to pressure changes in Figure 3b), leading to the formation of two new CO_2 accumulations. The advective CO_2 migration lasting for more than 200 years strongly dominates over diffusive transport. Nevertheless, the third caprock in the sequence cuts the percolating path, and molecular diffusion begins to govern CO_2 leakage. The diffusive CO_2 front rise is limited to 300 m after 1 million years.

A significant portion of the injected CO_2 initially exists in free phase (>50%) or dissolved in brine (>10%) outside the target aquifer (Figure 4b). The contribution of secondary traps to storage slightly decreases in the long term as more CO_2 diffuses upward. Remarkably, the CO_2 accumulation below the third caprock completely dissolves into brine after several hundred thousand years, leading to a drop in CO_2 concentration (Figure 3b). Furthermore, almost one-third of the mobile CO_2 within the reservoir becomes trapped by dissolution into underlying formations.

4. Discussion and Conclusion

The one-dimensional transport model developed in this study provides a high-resolution, computationally cost-effective, yet realistic assessment of CO_2 leakage from a gigatonne-scale storage scenario over geological time scales. The relatively fast advective flow may dominate CO_2 migration during and shortly after injection for caprocks with a low entry pressure and relatively high permeability. However, CO_2 does not migrate far up across the multi-layered system even if the caprocks are pervasively fractured. As the CO_2 pressure gradually drops, imbibition of brine into caprocks further hinders the CO_2 leakage. The CO_2 plume that migrates out of the target storage formation accumulates below the overlying caprocks forming secondary traps. Our observations agree well with field examinations of naturally occurring CO_2 in stacked reservoirs that have rarely been accompanied



Geophysical Research Letters



Figure 3. Dissolved CO_2 concentration profiles. Temporal evolution of the CO_2 concentration along a sequence of aquifers and caprocks for the (a) best-case and (b) worst-case leakage assessment scenarios.

by leakage to the surface over geological time scales (Miocic et al., 2016). In the long term, the diffusive leakage will not jeopardize storage security because the aqueous CO_2 can barely diffuse to a distance of 1 m over thousands of years. Our estimates of the diffusive migration rate pose an upper bound limit to previous predictions by reactive-diffusive transport modeling of a few millimeters to a few centimeters on a 1,000-year time scale (Gherardi et al., 2007; Kampman et al., 2016).

A single caprock would overall behave in a similar manner. CO_2 may create a long percolating path into the caprock as it does not anymore displace large brine volumes of alternating aquifers when forming secondary CO₂ accumulations. If CO₂ breaks through the caprock, a trade-off will likely arise between CO₂ dissolution in resident brine and buoyant rise (Sharma & Van Gent, 2018). CO₂ density decreases as it ascends, featuring its significant expansion at depths of around 800 m, where CO2 undergoes a phase transition to the gaseous state, favoring strong buoyant rise (Pruess, 2004). Thus, the possibility of CO₂ leakage back to the surface will be governed by the amount of leaked CO₂ from the caprock available for migration through the overlying high-permeability sediments. Taking the Span and Wagner (1996) equation of state to calculate CO₂ density and Henry's law of solubility with coefficients extracted for CO₂ from experimental data at a constant temperature of 40°C (Spycher et al., 2003), we argue that a CO_2 accumulation height of 35 m at a depth of 1,500 m is required to statically saturate the pore fluid of rock formations up to the surface (assuming a uniform porosity profile). At the same time, the self-limiting hydrodynamic mechanisms could appear to drastically attenuate leakage (Oldenburg & Unger, 2003; Pruess, 2004). Nonetheless, if injection in this type of stratigraphic setting is needed to unlock more storage capacity, a caprock of even low-sealing capacity but enough thickness should exist to ensure secure retention of CO_2 underground (see Section S2 in Supporting Information S1 for more details). In the meantime, adopting appropriate techniques to monitor the evolution of the free-phase CO₂ front in the more critical injection and early post-injection periods would be of paramount importance (Romanak et al., 2012; Vilarrasa et al., 2019; Zheng et al., 2022).





Figure 4. The long-term fate of the injected CO_2 underground. Temporal evolution of the CO_2 mass fractions for (a) the best-case and (b) the worst-case leakage assessment scenarios.

The isothermal conditions considered in our analysis overlook the temperature decline corresponding to the geothermal gradient as CO_2 migrates toward the surface. The temperature drop may be more pronounced due to the Joule-Thomson cooling effect, resulting from CO_2 expansion and depressurization following its phase transition from supercritical to gas at shallow depths (Vilarrasa & Rutqvist, 2017). However, the isothermal assumption would be conservative in CO_2 leakage assessment as the temperature decrease drives an increase in CO_2 viscosity (Altunin & Sakhabetdinov, 1972) and solubility in the resident brine (Spycher et al., 2003), as well as a reduction in the diffusion coefficient of CO_2 in water (Cadogan et al., 2014), all retarding CO_2 flow and transport through

19448007, 2022, 24, Downlo

loi/10.1029/2022GL100443 by Cochrane

Wiley Online Library on [23/04/2023]. See the Terms

s and Con

of use; OA

by the

applicable Creative

overlying formations. Yet, shallow sediments accommodate more CO₂ in free phase as a result of higher CO₂ density at lower temperatures (Span & Wagner, 1996). Thermally induced changes in CO₂ density and viscosity may affect its flow and, thus, alter pressure profiles, although such thermal effects are of secondary importance to CO₂ flow owing to the intrinsically slow heat transfer compared to hydraulic diffusivity of fluids (longer characteristic time of thermal vs. hydraulic processes, Taron et al., 2009).

By neglecting residual and mineral trapping mechanisms in the aquifer alongside transport-limited chemical reactions in the caprock, our evaluations of the vertical CO₂ migration are on the conservative side from the viewpoint of leakage potential. In contrast, the presence of a preferential flow conduit like a high permeability fault or fracture zone may violate our assumption of laterally uniform CO₂ leakage. Localized flow imposes minor changes in the reservoir pressure that may sustain a long-lived and vertically extended leakage (Oldenburg & Rinaldi, 2011). However, the knowledge of flow properties of faults in clay-rich formations is yet to be developed. Insights from recent experiments at the Mont Terri underground rock laboratory (Switzerland) point to significant shale fault heterogeneity with permeability variations, depending on the fault architecture, from 10^{-18} to 10^{-20} m² (Kneuker et al., 2017; Zappone et al., 2021). These values are comparable to those of Eau Claire shale or Opalinus Clay that are considered as caprock analogs (Kim & Makhnenko, 2020). The presence of flow heterogeneities and the ductile nature of clay-rich rocks, enabling fault reactivation with slight shear-induced dilation and permeability creation, are argued to considerably retard localized CO₂ migration up along the fault (Ikari et al., 2009; Rutqvist et al., 2016; Vilarrasa & Carrera, 2015). Indeed, significant permeability enhancement is anticipated after tensile fracturing occurring at notably high overpressure (Guglielmi et al., 2021), which is commonly avoided in the course of CO₂ injection. Besides, CO₂ flow through the fault may promote mineral precipitation (Nooraiepour et al., 2018) and dissolution-enhanced fine mobilization (Noiriel et al., 2007), resulting in self-sealing. Conclusively, although key research challenges concerning focused leakage (im)possibility through fault-like features have yet to be addressed, clay-rich caprocks most likely remain prominent seals for permanent gigatonne-scale CO₂ storage.

Data Availability Statement

The input files and the FEM code CODE_BRIGHT used for numerical simulations in this study are publicly available at the institutional repository Digital. CSIC, which practices FAIR principles (https://digital.csic.es/ handle/10261/261845).

References

- Alcalde, J., Flude, S., Wilkinson, M., Johnson, G., Edlmann, K., Bond, C. E., et al. (2018). Estimating geological CO2 storage security to deliver on climate mitigation. Nature Communications, 9, 1-13. https://doi.org/10.1038/s41467-018-04423-1
- Altunin, V. V., & Sakhabetdinov, M. A. (1972). Viscosity of liquid and gaseous carbon dioxide at temperatures 220-1300 K and pressure up to 120 bar. Teploenergetika, 8, 85-89.
- Bachu, S. (2003). Screening and ranking of sedimentary basins for sequestration of CO2 in geological media in response to climate change. Environmental Geology, 44(3), 277-289. https://doi.org/10.1007/s00254-003-0762-9
- Bandilla, K. W., & Celia, M. A. (2017). Active pressure management through brine production for basin-wide deployment of geologic carbon sequestration. International Journal of Greenhouse Gas Control, 61, 155-167. https://doi.org/10.1016/j.ijggc.2017.03.030
- Benson, S. B., & Cook, P. (2005). Underground geological storage. In Carbon Dioxide capture and storage (pp. 195-277). Cambridge University Press
- Bhuiyan, M. H., Agofack, N., Gawel, K. M., & Cerasi, P. R. (2020). Micro- and macroscale consequences of interactions between CO2 and shale rocks. Energies, 13(5), 1167. https://doi.org/10.3390/en13051167
- Blackford, J., Stahl, H., Bull, J. M., Bergès, B. J. P., Cevatoglu, M., Lichtschlag, A., et al. (2014). Detection and impacts of leakage from sub-seafloor deep geological carbon dioxide storage. Nature Climate Change, 4(11), 1011-1016. https://doi.org/10.1038/nclimate2381
- Boulin, P. F., Bretonnier, P., Vassil, V., Samouillet, A., Fleury, M., & Lombard, J.-M. (2013). Sealing efficiency of caprocks: Experimental investigation of entry pressure measurement methods. Marine and Petroleum Geology, 48, 20-30. https://doi.org/10.1016/j.marpetgeo.2013.07.010
- Busch, A., Alles, S., Gensterblum, Y., Prinz, D., Dewhurst, D. N., Raven, M. D., et al. (2008). Carbon dioxide storage potential of shales. International Journal of Greenhouse Gas Control, 2(3), 297-308. https://doi.org/10.1016/j.ijggc.2008.03.003
- Busch, A., Bertier, P., Gensterblum, Y., Rother, G., Spiers, C. J., Zhang, M., & Wentinck, H. M. (2016). On sorption and swelling of CO₂ in clays. Geomechanics and Geophysics for Geo-Energy and Geo-Resources, 2, 111-130. https://doi.org/10.1007/s40948-016-0024-4
- Cadogan, S. P., Maitland, G. C., & Trusler, J. M. (2014). Diffusion coefficients of CO2 and N2 in water at temperatures between 298.15 K and 423.15 K at pressures up to 45 MPa. Journal of Chemical & Engineering Data, 59(2), 519-525. https://doi.org/10.1021/je401008s
- Cavanagh, A. J., & Haszeldine, R. S. (2014). The Sleipner storage site: Capillary flow modelling of a layered CO2 plume requires fractured shale barriers within the Utsira formation. International Journal of Greenhouse Gas Control, 21, 101-112. https://doi.org/10.1016/j. ijggc.2013.11.017
- Dance, T., LaForce, T., Glubokovskikh, S., Ennis-King, J., & Pevzner, R. (2019). Illuminating the geology: Post-injection reservoir characterisation of the CO2CRC Otway site. International Journal of Greenhouse Gas Control, 86, 146–157. https://doi.org/10.1016/j.ijggc.2019.05.004

Acknowledgments

I.R.K. and V.V. acknowledge funding from the European Research Council (ERC) under the European Union's Horizon 2020 Research and Innovation Program through the Starting Grant GEoREST (www.georest.eu) under Grant agreement No. 801809. I.R.K. also acknowledges support by the PCI2021-122077-2B project (www. easygeocarbon.com) funded by MCIN/ AEI/10.13039/501100011033 and the European Union NextGenerationEU/ PRTR. IDAEA-CSIC is a Centre of Excellence Severo Ochoa (Spanish Ministry of Science and Innovation, Grant CEX2018-000794-S funded by MCIN/ AEI/10.13039/501100011033). R.Y.M. acknowledges the support from US DOE through Carbon SAFE Illinois Corridor Project DE-FE0031892. Additional funding for completing this manuscript was provided to C.M.O. and J.R. by the U.S. Department of Energy under contract No. DE-AC0205CH11231 to the Lawrence Berkeley National Laboratory.

Delkhahi, B., Nassery, H. R., Vilarrasa, V., Alijani, F., & Ayora, C. (2020). Impacts of natural CO₂ leakage on groundwater chemistry of aquifers from the Hamadan Province, Iran. *International Journal of Greenhouse Gas Control*, 96, 103001. https://doi.org/10.1016/j.ijgcc.2020.103001 De Simone, S., Jackson, S. J., & Krevor, S. (2019). The error in using superposition to estimate pressure during multisite subsurface CO₂ storage. *Geophysical Research Letters*, 46(12), 6525–6533. https://doi.org/10.1029/2019GL082738

Espinoza, D. N., & Santamarina, J. C. (2012). Clay interaction with liquid and supercritical CO₂: The relevance of electrical and capillary forces. International Journal of Greenhouse Gas Control, 10, 351–362. https://doi.org/10.1016/j.ijgcc.2012.06.020

Espinoza, D. N., & Santamarina, J. C. (2017). CO₂ breakthrough-caprock sealing efficiency and integrity for Carbon geological storage. International Journal of Greenhouse Gas Control, 66, 218–229. https://doi.org/10.1016/j.ijggc.2017.09.019

Farrar, C. D., Sorey, M. L., Evans, W. C., Howle, J. F., Kerr, B. D., Kennedy, B. M., et al. (1995). Forest-killing diffuse CO₂ emission at Mammoth Mountain as a sign of magmatic unrest. *Nature*, 376(6542), 675–678. https://doi.org/10.1038/376675a0

- Gaus, I., Azaroual, M., & Czernichowski-Lauriol, I. (2005). Reactive transport modelling of the impact of CO₂ injection on the clayey caprock at Sleipner (North Sea). *Chemical Geology*, 217(3–4), 319–337. https://doi.org/10.1016/j.chemgeo.2004.12.016
- Gherardi, F., Xu, T., & Pruess, K. (2007). Numerical modeling of self-limiting and self-enhancing caprock alteration induced by CO₂ storage in a depleted gas reservoir. *Chemical Geology*, 244(1–2), 103–129. https://doi.org/10.1016/j.chemgeo.2007.06.009 Global CCS Institute. (2021). Global status of CCS 2021.
- Guglielmi, Y., Nussbaum, C., Cappa, F., Barros, L. D., Rutqvist, J., & Birkholzer, J. (2021). Field-scale fault reactivation experiments by fluid injection highlight aseismic leakage in caprock analogs: Implications for CO₂ sequestration. *International Journal of Greenhouse Gas Control*, 111, 103471. https://doi.org/10.1016/j.ijgcc.2021.103471

Gunter, W. D., Bachu, S., & Benson, S. (2004). The role of hydrogeological and geochemical trapping in sedimentary basins for secure geological storage of carbon dioxide. *Geological Society - Special Publications*, 233(1), 129–145. https://doi.org/10.1144/GSL.SP.2004.233.01.09

- Hart, D. J., Bradbury, K. R., & Feinstein, D. T. (2006). The vertical hydraulic conductivity of an aquitard at two spatial scales. Ground Water, 44(2), 201–211. https://doi.org/10.1111/j.1745-6584.2005.00125.x
- Heath, J. E., Dewers, T. A., McPherson, B. J., Nemer, M. B., & Kotula, P. G. (2012). Pore-lining phases and capillary breakthrough pressure of mudstone caprocks: Sealing efficiency of geologic CO₂ storage sites. *International Journal of Greenhouse Gas Control*, 11, 204–220. https:// doi.org/10.1016/j.ijggc.2012.08.001
- Hepple, R. P., & Benson, S. M. (2005). Geologic storage of carbon dioxide as a climate change mitigation strategy: Performance requirements and the implications of surface seepage. *Environmental Geology*, 47(4), 576–585. https://doi.org/10.1007/s00254-004-1181-2
- Hildenbrand, A., Schlomer, S., & Krooss, B. M. (2002). Gas breakthrough experiments on finegrained sedimentary rocks. *Geofluids*, 2(1), 3–23. https://doi.org/10.1046/j.1468-8123.2002.00031.x
- Hill, P. M. (2000). Possible asphyxiation from carbon dioxide of a cross-country skier in eastern California: A deadly volcanic hazard. Wilderness and Environmental Medicine, 11(3), 192–195. https://doi.org/10.1580/1080-6032(2000)011[0192:PAFCDO]2.3.CO;2
- Hou, Z., Rockhold, M. L., & Murray, C. J. (2012). Evaluating the impact of caprock and reservoir properties on potential risk of CO₂ leakage after injection. *Environmental Earth Sciences*, 66(8), 2403–2415. https://doi.org/10.1007/s12665-011-1465-2
- Ikari, M. J., Saffer, D. M., & Marone, C. (2009). Frictional and hydrologic properties of clay-rich fault gouge. Journal of Geophysical Research, 114(B5), B05409. https://doi.org/10.1029/2008JB006089
- Intergovernmental Panel on Climate Change (IPCC). (2018). Summary for policymakers. In V. Masson-Delmotte, P. Zhai, H. O. Portner, D. Roberts, J. Skea, P. R. Shukla, et al. (Eds.), *Global warming of 1.5°C. An IPCC special report on the impacts of global warming of 1.5°C above pre-industrial levels and related global greenhouse gas emission pathways, in the context of strengthening the global response to the threat of climate change, sustainable development, and efforts to eradicate poverty (pp. 1–24). Intergovernmental Panel on Climate Change.*
- Juanes, R., MacMinn, C. W., & Szulczewski, M. L. (2010). The footprint of the CO₂ plume during carbon dioxide storage in saline aquifers: Storage efficiency for capillary trapping at the basin scale. *Transport in Porous Media*, 82(1), 19–30. https://doi.org/10.1007/s11242-009-9420-3
- Kampman, N., Busch, A., Bertier, P., Snippe, J., Hangx, S., Pipich, V., et al. (2016). Observational evidence confirms modelling of the long-term integrity of CO₂-reservoir caprocks. *Nature Communications*, 7(1), 12268. https://doi.org/10.1038/ncomms12268
- Kim, K., & Makhenko, R. (2020). Coupling between poromechanical behavior and fluid flow in tight rock. Transport in Porous Media, 135(2), 487–512. https://doi.org/10.1007/s11242-020-01484-z
- Kivi, I. R., Makhnenko, R. Y., & Vilarrasa, V. (2022). Two-phase flow mechanisms controlling CO₂ intrusion into Shaly caprock. Transport in Porous Media, 141(3), 771–798. https://doi.org/10.1007/s11242-022-01748-w
- Kivi, I. R., Vilarrasa, V., & Makhnenko, R. (2021). Effect of caprock relative permeability on CO₂ flow through it. In TCCS-11- trondheim conference on CO₂ capture, transport and storage. June 21–23.
- Kneuker, T., Hammer, J., Shao, H., Schuster, K., Furche, M., & Zulauf, G. (2017). Microstructure and composition of brittle faults in claystones of the Mont Terri rock laboratory (Switzerland): New data from petrographic studies, geophysical borehole logging and permeability tests. Engineering Geology, 231, 139–156. https://doi.org/10.1016/j.enggeo.2017.10.016
- Krevor, S. C. M., Pini, R., Li, B., & Benson, S. M. (2011). Capillary heterogeneity trapping of CO₂ in a sandstone rock at reservoir conditions. *Geophysical Research Letters*, 38(15), L15401. https://doi.org/10.1029/2011GL048239
- Lüth, S., Bergmann, P., Huang, F., Ivandic, M., Ivanova, A., Juhlin, C., & Kempka, T. (2017). 4D seismic monitoring of CO₂ storage during injection and post-closure at the Ketzin pilot site. *Energy Procedia*, 114, 5761–5767. https://doi.org/10.1016/j.egypro.2017.03.1714

MacMinn, C. W., Szulczewski, M. L., & Juanes, R. (2011). CO₂ migration in saline aquifers. Part 2. Capillary and solubility trapping. Journal of Fluid Mechanics, 688, 321–351. https://doi.org/10.1017/jfm.2011.379

- Makhnenko, R. Y., Vilarrasa, V., Mylnikov, D., & Laloui, L. (2017). Hydromechanical aspects of CO₂ breakthrough into Clay-rich caprock. *Energy Procedia*, 114, 3219–3228. https://doi.org/10.1016/j.egypro.2017.03.1453
- Miocic, J. M., Gilfillan, S. M. V., Roberts, J. J., Edlmann, K., McDermott, C. I., & Haszeldine, R. S. (2016). Controls on CO₂ storage security in natural reservoirs and implications for CO₂ storage site selection. *International Journal of Greenhouse Gas Control*, 51, 118–125. https://doi. org/10.1016/j.ijggc.2016.05.019
- Monastersky, R. (2013). Seabed scars raise questions over carbon-storage plan. *Nature*, 504(7480), 339–340. https://doi.org/10.1038/504339a
- Neuzil, C. E. (2019). Permeability of clays and shales. Annual Review of Earth and Planetary Sciences, 47(1), 247–273. https://doi.org/10.1146/ annurev-earth-053018-060437
- Noiriel, C., Madé, B., & Gouze, P. (2007). Impact of coating development on the hydraulic and transport properties in argillaceous limestone fracture. Water Resources Research, 43(9), W09046. https://doi.org/10.1029/2006WR005379
- Nooraiepour, M., Fazeli, H., Miri, R., & Hellevang, H. (2018). Effect of CO₂ phase states and flow rate on salt precipitation in shale caprocks—A microfluidic study. *Environmental Science and Technology*, 52(10), 6050–6060. https://doi.org/10.1021/acs.est.8b00251
- Nordbotten, J. M., Celia, M. A., & Bachu, S. (2005). Injection and storage of CO₂ in deep saline aquifers: analytical solution for CO₂ plume evolution during injection. *Transport in Porous Media*, 58(3), 339–360. https://doi.org/10.1007/s11242-004-0670-9

- Oldenburg, C. M. (2008). Screening and ranking framework for geologic CO₂ storage site selection on the basis of health, safety, and environmental risk. *Environmental Geology*, 54(8), 1687–1694. https://doi.org/10.1007/s00254-007-0947-8
- Oldenburg, C. M., & Rinaldi, A. P. (2011). Buoyancy effects on upward brine displacement caused by CO₂ injection. *Transport in Porous Media*, 87(2), 525–540. https://doi.org/10.1007/s11242-010-9699-0
- Oldenburg, C. M., & Unger, A. J. A. (2003). On leakage and seepage from geologic Carbon sequestration sites: Unsaturated zone attenuation. *Vadose Zone Journal*, 2(3), 287–296. https://doi.org/10.2136/vzj2003.2870
- Olivella, S., Gens, A., Carrera, J., & Alonso, E. E. (1996). Numerical formulation for a simulator (CODE BRIGHT) for the coupled analysis of saline media. *Engineering Computations*, 13(7), 87–112. https://doi.org/10.1108/02644409610151575
- Pawar, R. J., Chu, S., Makedonska, N., Onishi, T., & Harp, D. (2020). Assessment of relationship between post-injection plume migration and leakage risks at geologic CO₂ storage sites. *International Journal of Greenhouse Gas Control*, 101, 103138. https://doi.org/10.1016/j. ijggc.2020.103138
- Pruess, K. (2004). Numerical simulation of CO₂ leakage from a geologic disposal reservoir, including transitions from super-to sub-critical conditions, and boiling of liquid CO₂. SPE Journal, 9(02), 237–248. https://doi.org/10.2118/86098-PA
- Rezaee, R., Saeedi, A., Iglauer, S., & Evans, B. (2017). Shale alteration after exposure to supercritical CO₂. International Journal of Greenhouse Gas Control, 62, 91–99. https://doi.org/10.1016/j.ijggc.2017.04.004
- Rezaeyan, A., Pipich, V., Ma, J., Leu, L., Seemann, T., Rother, G., et al. (2022). Predicting fluid flow regime, permeability, and diffusivity in mudrocks from multiscale pore characterisation. *Transport in Porous Media*, 141(1), 201–229. https://doi.org/10.1007/s11242-021-01717-9
- Romanak, K. D., Bennett, P. C., Yang, C., & Hovorka, S. D. (2012). Process-based approach to CO₂ leakage detection by vadose zone gas monitoring at geologic CO₂ storage sites. *Geophysical Research Letters*, 39(15), L15405. https://doi.org/10.1029/2012GL052426
- Rutqvist, J. (2012). The geomechanics of CO₂ storage in deep sedimentary formations. *Geotechnical & Geological Engineering*, 30(3), 525–551. https://doi.org/10.1007/s10706-011-9491-0
- Rutqvist, J., Rinaldi, A. P., Cappa, F., Jeanne, P., Mazzoldi, A., Urpi, L., et al. (2016). Fault activation and induced seismicity in geological carbon storage—Lessons learned from recent modeling studies. *Journal of Rock Mechanics and Geotechnical Engineering*, 8(6), 789–804. https:// doi.org/10.1016/j.jrmge.2016.09.001
- Scott, V., Gilfillan, S., Markusson, N., Chalmers, H., & Haszeldine, R. S. (2012). Last chance for carbon capture and storage. Nature Climate Change, 3(2), 105–111. https://doi.org/10.1038/nclimate1695
- Sharma, S., & Van Gent, D. (2018). The Australian South West hub project: Developing confidence in migration assisted trapping in a saline aquifer – Understanding uncertainty boundaries through scenarios that stress the models. In 14th greenhouse gas control technologies conference. GHGT-14. October 21–26.
- Song, J., & Zhang, D. (2013). Comprehensive review of caprock-sealing mechanisms for geologic carbon sequestration. *Environmental Science and Technology*, 47(1), 9–22. https://doi.org/10.1021/es301610p
- Span, R., & Wagner, W. (1996). A new equation of state for carbon dioxide covering the fluid region from the triple point temperature to 1100 K at pressures up to 800 MPa. Journal of Physical and Chemical Reference Data, 25(6), 1509–1596. https://doi.org/10.1063/1.555991
- Spycher, N., Pruess, K., & Ennis-King, J. (2003). CO₂-H₂O mixtures in the geological sequestration of CO₂. I. Assessment and calculation of mutual solubilities from 12 to 100°C and up to 600 bar. *Geochimica et Cosmochimica Acta*, 67(16), 3015–3031. https://doi.org/10.1016/ S0016-7037(03)00273-4
- Taron, J., Elsworth, D., & Min, K. B. (2009). Numerical simulation of thermal-hydrologic-mechanical-chemical processes in deformable, fractured porous media. International Journal of Rock Mechanics and Mining Sciences, 46(5), 842–854. https://doi.org/10.1016/j.ijrmms.2009.01.008
- Tewes, F., & Boury, F. (2005). Formation and rheological properties of the supercritical CO₂–Water pure interface. *The Journal of Physical Chemistry B*, 109(9), 3990–3997. https://doi.org/10.1021/jp046019w
- Vilarrasa, V. (2022). Numerical models for the simulation of potential CO₂ leakage through Shaly caprocks over geological time scales [Dataset]. Institutional repository DIGITAL. CSIC. Retrieved from https://digital.csic.es/handle/10261/261845
- Vilarrasa, V., Bolster, D., Olivella, S., & Carrera, J. (2010). Coupled hydromechanical modeling of CO₂ sequestration in deep saline aquifers. International Journal of Greenhouse Gas Control, 4(6), 910–919. https://doi.org/10.1016/j.ijggc.2010.06.006
- Vilarrasa, V., & Carrera, J. (2015). Geologic carbon storage is unlikely to trigger large earthquakes and reactivate faults through which CO₂ could leak. *Proceedings of the National Academy of Sciences of the United States of America*, 112, 5938–5943. https://doi.org/10.1073/pnas.1413284112
- Vilarrasa, V., Carrera, J., Olivella, S., Rutqvist, J., & Laloui, L. (2019). Induced seismicity in geologic Carbon storage. Solid Earth, 10(3), 871–892. https://doi.org/10.5194/se-10-871-2019
- Vilarrasa, V., Makhnenko, R., & Gheibi, S. (2016). Geomechanical analysis of the influence of CO₂ injection location on fault stability. *Journal of Rock Mechanics and Geotechnical Engineering*, 8(6), 805–818. https://doi.org/10.1016/j.jrmge.2016.06.006
- Vilarrasa, V., Olivella, S., Carrera, J., & Rutqvist, J. (2014). Long term impacts of cold CO₂ injection on the caprock integrity. *International Journal of Greenhouse Gas Control*, 24, 1–13. https://doi.org/10.1016/j.ijgcc.2014.02.016
- Vilarrasa, V., & Rutqvist, J. (2017). Thermal effects on geologic carbon storage. *Earth-Science Reviews*, 165, 245–256. https://doi.org/10.1016/j.earscirev.2016.12.011
- Zahasky, C., & Benson, A. M. (2016). Evaluation of hydraulic controls for leakage intervention in carbon storage reservoirs. International Journal of Greenhouse Gas Control, 47, 86–100. https://doi.org/10.1016/j.ijgcc.2016.01.035
- Zapata, Y., Kristensen, M. R., Huerta, N., Brown, C., Kabir, C. S., & Reza, Z. (2020). CO₂ geological storage: Critical insights on plume dynamics and storage efficiency during long-term injection and post-injection periods. *Journal of Natural Gas Science and Engineering*, 83, 103542. https://doi.org/10.1016/j.jngse.2020.103542
- Zappone, A., Rinaldi, A. P., Grab, M., Wenning, Q. C., Roques, C., Madonna, C., et al. (2021). Fault sealing and caprock integrity for CO₂ storage: An in situ injection experiment. *Solid Earth*, *12*(2), 319–343. https://doi.org/10.5194/se-12-319-2021
- Zhao, X., Deng, H., Wang, W., Han, F., Li, C., Zhang, H., & Dai, Z. (2017). Impact of naturally leaking carbon dioxide on soil properties and ecosystems in the Qinghai-Tibet Plateau. Scientific Reports, 7(1), 3001. https://doi.org/10.1038/s41598-017-02500-x
- Zheng, X., Espinoza, D. N., Vandamme, M., & Pereira, J. M. (2022). CO₂ plume and pressure monitoring through pressure sensors above the caprock. *International Journal of Greenhouse Gas Control*, 117, 103660. https://doi.org/10.1016/j.ijggc.2022.103660
- Zhou, Q., Birkholzer, J. T., Mehnert, E., Lin, Y. F., & Zhang, K. (2010). Modeling basin- and plume-scale processes of CO₂ storage for full-scale deployment. Ground Water, 48(4), 494–514. https://doi.org/10.1111/j.1745-6584.2009.00657.x

References From the Supporting Information

Rutqvist, J., Vasco, D. W., & Myer, L. (2010). Coupled reservoir-geomechanical analysis of CO₂ injection and ground deformations at in Salah, Algeria. International Journal of Greenhouse Gas Control, 4(2), 225–230. https://doi.org/10.1016/j.ijgc.2009.10.017

Van Genuchten, M. T. (1980). A closed-form equation for predicting the hydraulic conductivity of unsaturated soils. Soil Science Society of America Journal, 44(5), 892–898. https://doi.org/10.2136/sssaj1980.03615995004400050002x